food & nutrition (

ORIGINAL ARTICLE

Changes in dietary habits and BMI *z*-score after a 6-month non-randomized cluster-controlled trial among 6–12 years old overweight and obese Norwegian children

Tonje Holte Stea¹*, Mario Vianna Vettore¹, Bente Øvrebø² and Eirik Abildsnes³

¹Department of Health and Nursing Science, University of Agder, Kristiansand, Norway; ²Department of Health and Inequalities, Norwegian Institute of Public Health, Oslo, Norway; ³Department of Psychosocial Health, University of Agder, Grimstad, Norway

Popular scientific summary

- A 6-month family-based intervention study was tailored to improve lifestyle habits among overweight and obese children in primary health settings.
- The intervention resulted in improved eating behaviors and reduced energy intake.
- A decrease in BMI z-score was observed in the intervention group compared to the control group after the intervention period.
- This study provides support for implementing complex intervention programs tailored to local primary care settings.

Abstract

Background: Effective prevention programs to address the high prevalence of childhood overweight and obesity and the concomitant health consequences have been warranted.

Objective: To improve dietary habits and weight status among Norwegian children with overweight/obesity in the primary care setting.

Design: A 6-month non-randomized cluster-controlled intervention among 137 children, aged 6–12 years, with overweight/obesity and their parents. Intervention and control groups were recruited by public health nurses and followed-up by 12 Healthy Life Centers across Norway. The intervention group received individual family counseling and participated in nutrition courses and physical activity groups. A frequency question-naire assessing sociodemographic characteristics and dietary habits was completed by the parents. Trained public health nurses measured height and weight using standardized methods to calculate body mass index (BMI) and BMI *z*-scores.

Results: The intervention resulted in an increased odds of consuming evening meals (OR: 3.42), a decreased availability of salty snacks ($\beta = -0.17$), a decreased intake of salty snacks (-0.18), an increased consumption of water ($\beta = 0.20$), and a decreased estimated total intake of energy ($\beta = -0.17$), carbohydrates ($\beta = -0.17$), mono- and disaccharides ($\beta = -0.21$), sucrose ($\beta = -0.24$), and saturated fatty acids ($\beta = -0.17$). The intervention directly predicted lower BMI *z*-score ($\beta = -0.17$), and post-treatment levels of energy ($\beta = -0.65$), saturated fat ($\beta = 0.43$), and total carbohydrates ($\beta = 0.41$) were directly linked to BMI *z*-score after intervention. Age and sex were indirectly associated with BMI after intervention through energy and saturated fat intake.

Conclusions: The intervention had a beneficial impact on nutrient intake and weight status among children with overweight/obesity. These findings provide support for implementing complex intervention programs tailored to local primary care settings.

Trial registration: Clinicaltrials.gov, NCT02290171. Registered 13. November 2014, https://clinicaltrials.gov/ ct2/show/NCT02290171

Keywords: overweight; obesity; children; family involvement; dietary intervention

Received: 20 April 2023; Revised: 3 November 2023; Accepted: 7 November 2023; Published: 12 December 2023

uring the past decades, the prevalence of childhood overweight and obesity has increased worldwide (1), but evidence suggests a stabilization in the prevalence of overweight and obesity in developed countries (2). Stabilization of overweight and obesity prevalence is also supported by results from a Norwegian longitudinal study among 8- to 13-year-olds, which reported that 16% of the adolescents were overweight and 3% were obese, and that the prevalence of overweight and obesity in adolescence seemed to be established during childhood (3). Childhood overweight and obesity often persists into adulthood and may lead to an increased risk of developing chronic disorders during early adulthood (4,5,6). Childhood obesity should be classified as a chronic disease in itself according to the Childhood Obesity Task Force (COTF) of the European Association for the Study of Obesity (EASO) (7). Such approach aims to increase the individual and societal awareness related to childhood obesity and improve early diagnosis and intervention (7). Nevertheless, type 2 diabetes, hypertension, dyslipidemia, metabolic syndrome, and obstructive sleep apnea are adverse health outcomes associated with overweight in children and adolescents (8), and the aforementioned risk factors might result in premature heart disease, cancer, and adult diabetes (6).

The link between overweight/obesity and poor health among children is complex due to the interrelationships among genetic, biological, behavioral, social, cultural, and environmental influences (9, 10). The main postulated explanations for the childhood obesity epidemic are unfavorable dietary habits, increased sedentary behavior, and reduced physical activity, and the impact of these risk behaviors on weight status is modified by age and sex (11). Findings from a study among Norwegian 11-year-olds indicated sex differences in behavioral correlates of weight status, but not for weight status itself (12). In a sample of Australian children, results also indicated changes in environmental factors with age, and that family food environment among older children was associated with children's body mass index (BMI) z-score and weight status cross-sectionally and longitudinally (13).

In addition, social inequalities related to body composition are well established, with greater fat mass, lower fat-free mass, and higher prevalence of overweight and obesity among disadvantaged children (14, 15). Studies have confirmed that diet quality may explain this relationship as children in the higher socioeconomic categories consume a healthier diet (16, 17) and have a healthier body weight (18). A study among Norwegian children has also confirmed socioeconomic inequalities in children's weight, height, and BMI trajectories (19). Moreover, findings from a study among children aged 6–10 years in the US suggested that the increase in food responsiveness and emotional overeating in obese children was influenced by maternal education (20).

A recent review has reported that first-line treatment approaches to tackle childhood obesity include family-based behavioral interventions focusing on improving healthy lifestyle habits, including dietary habits, physical activity, sedentary behaviors, and sleep quality, underpinned by behavior change strategies (21). In Norway, measurements of height and weight among all children entering 1st, 3rd, and 8th grade are routinely collected every year at school by public health nurses. According to the national health authorities, public health nurses are obliged to measure and report childhood overweight and obesity to parents and provide further advice on weight management (22). However, national guidelines do not provide specific recommendations or practical guidelines for the prevention and management of overweight and obesity in children (23).

A previous systematic review has emphasized that in general, national screening programs aiming to identify children with overweight/obesity are not justified without follow-up effective intervention programs (24). However, findings from most of the published dietary intervention studies aiming to prevent or reduce childhood obesity have shown no or limited effects on weight reductions (25). On the other hand, another systematic review concluded that complex lifestyle interventions targeting children and adolescents with overweight/obesity and involving structured strategies for change in diet and physical activity may reduce BMI and BMI z-score compared with less complex interventions (26). A Cochrane review also confirmed that multi-component behavior-changing interventions may be beneficial in achieving small, short-term reductions in BMI and BMI z-score in children aged 6-11 years, but the overall quality of the evidence was low or very low partly due to differences in study design, setting, parental involvement, and inconsistent results (27). The ecological model for health promotion has highlighted the importance of the influence of both individual and environmental factors, including parent-child interaction, when planning health promotion interventions for children (28). Although family-based behavioral programs, which specifically target parents as the exclusive mediator of obesity treatment of children, are known to be an effective approach to reduce childhood obesity, few studies have included parents as behavioral agents (29). Due to the lack of high-quality evidence, further research is warranted in which health practitioners can work with parents as agents of change and focus on fostering positive parenting skills in order to support health behaviors in their children (30).

Our research group has previously published results from a systematic and evidence-based pilot study developed to specifically target children with overweight/obesity and their families with program components tailored to the local context in Norwegian municipalities (31). The theoretical model that we have developed hypothesized that the intervention, child's age, and sex would be directly associated with nutrients and BMI *z*-score after the intervention. Indirect effects of age, sex, and the intervention on BMI *z*-score after intervention mediated by nutrients were also hypothesized. Based on findings from the pilot study, we developed a family-based, multi-component program to improve healthy lifestyle habits among children and adolescents in a primary care setting. The main aim of this present study was to examine possible effects of the developed intervention on dietary habits and weight status among children with overweight and obesity.

Materials and methods

Study design and participants

This 6-month non-randomized cluster-controlled trial was based on the results of a 1-year pilot study, which was systematically developed and implemented as a tailored family-based intervention to improve lifestyle habits, by encouraging physical activity and healthy dietary habits, enhancing parental self-efficacy, and enhancing family engagement and parent-child relationships (31). In the present study, we evaluated the possible effects of the 6-month intervention on meal pattern, food and beverage availability and consumption, nutrient intake, and weight status among children with overweight/ obesity.

The study population consisted of children with overweight/obesity and their families, recruited from October 2014 to September 2016 by public health nurses after routine screening of weight and height in preschoolers and among 3rd grade pupils. The International Obesity Task Force (IOTF) BMI cut-offs and IOTF LMS parameters were used to categorize BMI and calculate BMI z-score, respectively (32). Healthy Life Centers (HLCs) and Public Health Clinics (PHCs) in the municipalities Bergen (two districts), Sola, Sandnes, Stavanger, Søgne, Kristiansand (three districts), Grimstad, Horten, and Oslo (two districts) were responsible for recruiting families and conducting the intervention in collaboration with local sports clubs. The families participating in the present study were recruited based on assessment of overweight and obesity after standard measurement of children's height and weight during school hours in 1st and 3rd grades. The recruitment process in regular services in several different municipalities did not allow the identification of the total number of children and families invited.

The intervention group and the control group were selected without any random pre-selection processes. HLCs and PHCs with an existing infrastructure facilitating the implementation of the intervention program recruited participants to the intervention group, whereas other HLCs and PHCs recruited participants to the control group. All participants in the control group were informed that they would receive the intervention program by HCLs and PHCs in their local community after serving as a control group for 6 months. Thus, the control group served as waiting list group and did not receive any counseling during the intervention period.

A total of 164 children and adolescents between 6 and 12 years old and their families agreed to participate in this study (control group: n = 73, intervention group: n = 90). Of them, a total of 77 participants from the intervention group and 66 participants from the control group completed the intervention period. Reasons for loss to follow-up included lack of time, moved away, conflicting time schedules, and participation in other leisure time activities.

Based on results from a previous intervention study targeting obese children for the promotion of healthy lifestyle habits and weight reduction, we conducted sample size calculations (33). The final sample of 143 participants, to detect effect size at least 0.15 (medium effect size) in a multiple regression analysis involving six independent variables with 5% Type I error probability, would lend a power of 94% (34).

Prior to participation in the present study, the children and their parents received oral and written information about the study. Parents signed a written consent form before completing an online questionnaire providing information on their children's lifestyle behaviors as well as demographic and sociodemographic data (e.g. sex, age, and educational attainment). Results from each participant were registered with a unique identification code provided by the local HCLs, which made it possible to link data collected during the intervention period and ensure anonymity, as the identity of individual participants was not known to the researchers. The Regional Committee for Medical and Health Research Ethics approved the study (no. 2013/1291), which was conducted in accordance with the Helsinki Declaration of 1975, as revised in 2008.

Deviation from the protocol

Originally, we intended to perform a cluster randomized clinical trial among overweight and obese children between 6 and 10 years old and their parents. However, the present intervention study used a quasi-experimental controlled design, as it was only feasible to recruit participants to the intervention group and the control group without any random pre-selection process. For ethical reasons, we also used a waiting list design, which did not allow long-term controlled follow-up as the control group received the same intervention as the intervention group with a 6-month delay. Finally, we decided to include a wider age range than described in the protocol, as the municipal healthcare institutions included in the present study had focus on screening and treatment of overweight and obesity in children between 6 and 12 years old.

The intervention program

Results from the systematic and evidence-based approach used for the development of this family-based intervention study have been described in the study protocol (31). We decided to focus on changing three key behaviors, i.e. level of physical activity, diet, and self-regulation skills. Furthermore, these behaviors were translated into 10 main program objectives: 1) increase level of physical activity, 2) decrease sedentary time and screen time viewing, 3) establishing regular meal patterns, 4) increasing number of family meals, 5) regulate/decrease portion sizes, 6) increase intake of healthy food and beverages and decrease intake of unhealthy alternatives, 7) establish adequate duration of sleep, 8) increase self-esteem, 9) strengthen autonomy support (parents), and 10) strengthen autonomous regulation and self-efficacy (parents). In addition, two subsidiary program objectives were described: 1) reducing BMI z-score and 2) improve self-perceived health.

Based on experience from the pilot study, we aimed at developing an intervention tailored to individual needs, enhancing parental motivation, and taking environmental factors into account. Thus, an ecological approach was used when developing the intervention, but the Self-Determination Theory (SDT) served as the main theoretical framework (35). According to SDT, different types of intrinsic and extrinsic motivation can underlie one's behavior. Intrinsic motivation refers to the inherent satisfaction related to the behavior, which was the central component of the motivational interview (MI) to enhance distinct attributes of the participants, such as empowerment and self-efficacy. Extrinsic motivation is defined as engaging in a behavior for instrumental reasons or to obtain some outcome disassociated from the behavior per se. It may include some motivation of personal value, including exercising to maintain good health, such as the physical activity sessions of the intervention. All participants in the intervention group received tailored individual (faceto-face) counseling and follow-up by certified health personnel using MI as a counseling technique in consultations to promote internal motivation, empowerment, and mastery of health (36). During the first meeting, an individual plan (IP) defining a maximum of three main goals for behavioral change followed by several more specific sub goals in the IP for each family during the intervention period was created in collaboration between parents and health personnel. A minimum of three and a maximum of eight individual consultations were offered to support all participating families in reaching these goals during the intervention period. Results from the previously published pilot study indicated that both parents and health

personnel responsible for the individual consultations were satisfied with the scope of the IP, the frequency of the individual consultations, and the use of MI as a technique designed to change specific health behaviors (31).

In addition to individual counseling and follow-up, parents and their children attended four-five face-to-face courses, developed by a nutrition scientist that included both theoretical and practical learnings sessions. All courses were tailored to meet the specific needs and challenges regarding nutritional knowledge, attitude, skills, self-efficacy, and intentions toward improving healthy dietary habits among this target group. Methods and strategies used during individual counseling and groupbased courses for the promotion of healthy eating also focused on changing home environment by mobilizing social support, provide positive reinforcement, and focus on the importance of high availability of healthy food choices and low availability of unhealthy food and beverages. In addition, children in the intervention group participated in physical activity sessions on a weekly basis (1-2 times per week), aiming at increasing moderate-to-vigorous intensity physical activity (MVPA) and improving motor control. The physical activity program was developed by physical activity researchers, who were experienced in planning, implementing, and evaluating the effect of physical activity sessions targeting children with overweight and obesity. Detailed information about program components and materials has been described in detail elsewhere (31). Intervention adherence and fidelity according to the protocol were controlled and approved for all participants by health personnel responsible for the individual consultations and group-based courses, and certified activity leaders responsible for physical activity groups.

Data collection

Weight and height were measured at baseline and after the 6-month intervention period by trained health professionals at the participating HLCs and PHCs. Information about sociodemographic background, meal pattern, and food and beverage availability and consumption were provided by parents completing online questionnaires using Survey Exact. Each participant was given a personal identification-number used to identify and link participants' data collected in the study.

Questionnaire

A self-administered food frequency questionnaire (FFQ) based on a Danish FFQ for children and adolescents was filled out by all parents at baseline and after a 6-month intervention period. The FFQ has been validated prior to this study (37) and consisted of 146 questions on the average consumption of food and beverage items during the past 4 weeks. These food and beverage items were divided

into different categories (i.e. beverage and dairy products, bread and cereals, spread/cold cuts, fish and meat, dinner meals, side dishes, fruit and vegetables, dessert, cakes, and snacks).

Information regarding intake of food items, dinner meals, side dishes, and beverages was assessed with the following question: How often does your child consume the selected food item/dinner meal/side dish/beverage? For specific food items, each question had seven response options: never, 1-3 times per month, once a week, 2-3 times per week, 4–6 times per week, and once or more times a day. Dinner meals and side dishes used the interval ranges: never, 1-3 per month, 1 per week, 2-4 per week, and more than 4 times per week. For the consumption of beverages, the interval ranges were never, 1-3 glasses per month, 1 glass per week, 2–6 glasses per week, 1 glass per day, 2-3 glasses per day, and more than 3 glasses per day. The respondents reported their consumption of food items, dinner meals, side dishes, and beverages in 'units per month', 'units per week', or 'units per day'. The unit measurements differed between items, but most questions were related to a standard portion size (e.g. cup of coffee, a piece of bread, and an apple). For some questions, extra information was provided (e.g. maize=2 tablespoons or soda = 0.5 L).

When calculating nutrient intake for each participant, the standard portion included in the FFQ (e.g. glass of milk), which was coded according to the Norwegian standards (38) in grams, was multiplied by the intake frequency registered in the FFQ. In addition, each food and beverage item had a food-code linkage corresponding to the numbers in the Norwegian food composition table. For dishes, we adjusted for estimated nutrient loss as a result of using different cooking procedures. Finally, we used FOOD-Calc and the Norwegian food composition table (39, 40) to calculate total nutrient intake for each participant at baseline and 6-month follow-up. In the present study, we reported total energy intake (kJ), total intake of proteins (g), and intake of carbohydrates (g), including total carbohydrate intake (g) and intakes of starch, mono-, and disaccharides (g), sucrose (g), and fiber (g/1,000 MJ). We also reported intake of fat, including total fat intake, and intakes of cholesterol (mg), saturated fatty acids (SFAs) (g), monounsaturated fatty acids (MUFAs) (g), polyunsaturated fatty acids (PUFAs) (g), n-3 fatty acids (g), and n-6 fatty acids (g). Finally, we present the percentage of total daily energy intake (E%) for protein, total fat, SFAs, MUFAs, PUFAs, total carbohydrate, and sucrose.

Information about the children's habitual consumption of main meals and family meals was obtained by asking respondents the following question: *How often do you usually eat breakfast, lunch, dinner, and evening and family meals each week?* Each question had five response options: never or seldom, 1–2 times per week, 3–4 times per week, 5–6 times per week, and daily. For statistical analyses, those who reported a daily intake of these main meals or having family meals daily were classified as regular breakfast-, lunch-, dinner-, and evening and family meal consumers, whereas those who omitted these main meals and family meal at least once a week were classified as irregular (family) meal consumers.

Information regarding food and beverage availability (availability of candy, salty snacks, diet-soda, sugar-sweetened soda and fruits and vegetables) was assessed using the following question: *How often are the selected food item/ beverage available?* Each question had 10 response options: *never, less than once a month, less than once a week, once a week, two times a week ... up to seven days a week.* In the statistical analysis, they were scored 0, 0.1, 0.5, 1, 2, 3 ... 7, to reflect the number of days per week these food and beverages were available for consumption at home.

The questionnaire also included sociodemographic information, such as child's age and sex, and parental education. Parental education level was assessed using the following question: What level of education do you have, answer for yourself and your partner. Both questions (for yourself and your partner) had six response options: elementary school <7 years, elementary school 7–10 years, vocational school/high school <3 years, high school -3 years, 3 years of high school, college or university ≤4 years, and colleges or university ≥ 4 years. These response alternatives were then trichotomized to reflect the following educational level for both parents: primary school, high school, and college/university. Parental education was included in statistical models as previous systematic reviews have reported an association between low socioeconomic position and higher levels of obesity in high-income countries, especially when parental education was used as a measure of socioeconomic position (41).

BMI and BMI z-scores

Trained public health nurses were responsible for weight and height measurements using standard calibrated scales and stadiometers to determine height to the nearest 0.1 cm and weight to the nearest 0.1 kg. All children were weighted wearing light clothing and without shoes. BMI was calculated (body mass (kg)/height² (m²)), and BMI *z*-scores were assigned to each participant (32).

Statistical analysis

Difference in prevalence of overweight and obesity and difference in sociodemographic characteristics between the intervention and the control group at baseline were analyzed using the independent sample t test for the continuous variable (age) and the Chi-Square Test for categorical variables. In this study, five outcomes were considered to evaluate the impact of the behavior

intervention, considering the control group as reference category: 1) meal patterns (categorical variables according to the frequency of daily breakfast, lunch, dinner, and evening and family meals), 2) availability of selected food and beverages (continuous variables based on the number of days per week of sugar-sweetened soda, diet soda, fruit and vegetables, sweets, and salty snacks), 3) consumption of food and beverages (continuous variables (g/day) according to the amount of different food and beverages intake), 4) BMI (continuous variables using kg/m² and z-score), and 5) nutrient intake (continuous variables of intake of energy (kJ) and other different measures in g and E%). Differences in frequency of meal consumption between the intervention group and the control group at baseline and after the intervention period were compared using the Chi-Square Test. The effect of the intervention on different meal patterns was tested by using multi-level logistic regression models. Significance of differences between the intervention group and the control group at baseline and after the intervention in mean counts of food and beverage availability and consumption, nutrient intake, BMI, and BMI z-score period was analyzed using the independent sample t-tests. The effect of the intervention on food and beverage availability, consumption, estimated nutrient intake, BMI, and BMI z-score was tested by using multiple multi-level linear regression models. The multi-level structure of analysis including 143 participants (level 1) grouped into the 12 HLCs and PHCs (level 2) was initially tested. A two-level random intercepts and fixed-slopes null model with individuals nested within HLCs and PHCs were fitted using the maximum likelihood method to assess the variance and standard error for BMI z-score after intervention across HLCs and PHCs (random effects). The variation of BMI z-scores after intervention at HLCs and PHCs level in the null multilevel model was not statistically significant when the

variance and standard error for the outcome at HLCs and PHCs level (random effects) were tested (Wald Z test = 0.971, P = 0.332). Therefore, multi-level analysis accounting for HLCs and PHCs level was not used. All models were adjusted for baseline scores, age, sex, and maternal and paternal education.

Path analysis was used to test the direct and indirect relationships between the intervention, child's age and sex, nutrients, BMI z-score at baseline, and BMI z-score post-treatment, according to the theoretical model (Fig. 1). The standardized direct effects refer to a direct path from one variable to another, whereas indirect effects are detected when one variable was the pathway between two other variables. The theoretical model hypothesized that the intervention and child's age and sex would be directly associated with nutrients and BMI z-score after the intervention. Indirect effects of age, sex, and the intervention on BMI z-score after intervention mediated by nutrients were also hypothesized. Initially, the full model using the maximum likelihood method was tested for identification. Then, non-substantial direct paths between variables (P-value > 0.10) were removed, and the parsimonious model was re-estimated to obtain standardized effects, standard errors, and P-values. The overall model fit of the full and parsimonious models was considered adequate according to the Comparative Fit Index (CFI) ≥ 0.90 (42). Statistical analyses were performed using the IBM SPSS version 25 and Stata program version 17 (StataCorp LP, College Station, TX, USA). For all analysis, P-value <0.05 was considered as statistically significant.

Results

Participant characteristics

According to results presented in Table 1, no differences in age, sex, and maternal education were shown at baseline



Fig. 1. Full theoretical model on the relationships between sex, age, intervention, nutrients, and BMI *z*-score. BMI: body mass index.

Table 1. Sociodemographic characteristics of the intervention and control group

| Sociodemographic characteristics | Intervention group ($n = 77$) | Control group $(n = 66)$ | P* |
|----------------------------------|---------------------------------|--------------------------|-------|
| Age (years) | 8.6 (8.1, 9.0) | 8.1 (7.8, 8.4) | 0.067 |
| Sex (girls) | 41 (59) | 35 (59) | 0.081 |
| Maternal educational level | | | |
| Primary scool | 6 (8.7) | 3 (5.2) | 0.461 |
| High school | 31 (44.9) | 32 (55.1) | |
| College/University | 32 (46.4) | 23 (39.7 | |
| Paternal educational level | | | |
| Primary scool | 8 (11.6) | 9 (16.1) | 0.041 |
| High school | 30 (43.5) | 34 (60.7) | |
| College/University | 31 (44.9) | 13 (23.2) | |

*Tested using independent sample t-test for continous variable (age) and chi-square test for categorical variables.

| Table 2. | Changes in meal | pattern during | the 6-months | intervention period |
|----------|-----------------|----------------|--------------|---------------------|
|----------|-----------------|----------------|--------------|---------------------|

| Daily meals | Ba | aseline | | 6 months | s of follow-u | ıp | Within-gro | up proporti | on changes after 6 mo | onths |
|---------------------|-----------------------|------------------|-------|-----------------------|------------------|-------|-----------------------|------------------|-----------------------|-------|
| | Intervention group | Control group | P* | Intervention group | Control group | P* | Intervention group | Control group | OR (95% CI) | P# |
| Breakfast, n (%) | 49 (71) | 44 (75) | 0.652 | 54 (78) | 42 (71) | 0.357 | 5 (7) | -2 (-4) | 2.45 (0.79, 7.58) | 0.120 |
| Lunch, n (%) | 52 (75) | 44 (75) | 0.918 | 52 (75) | 46 (78) | 0.729 | 0 (0) | 2 (3) | 0.59 (0.21, 1.66) | 0.313 |
| Dinner, n (%) | 56 (81) | 51 (86) | 0.421 | 61 (88) | 49 (83) | 0.385 | 5 (7) | -2 (3) | 1.27 (0.34, 4.79) | 0.720 |
| Evening meal, n (%) | 35 (51) | 37 (63) | 0.173 | 47 (68) | 31 (53) | 0.072 | 10 (17) | -6 (-10) | 3.42 (1.21, 9.66) | 0.020 |
| Family meals, n (%) | 39 (57) | 34 (58) | 0.900 | 42 (61) | 35 (59) | 0.857 | 3 (4) | 1(1) | 1.29 (0.52, 3.25) | 0.580 |

*Difference in meal pattern between the intervention group and the control group was analyzed using Chi-square tests.

"Intervention effect analyzed as difference in mean scores at baseline and after 6 months of follow-up in each variable (of each group) adjusted for baseline scores, age, sex, maternal- and paternal educational level using logistic regression models.

CI: confidence intervals; OR: odds ratio.

p-value < 0.05 was considered as statistically significant.

between the intervention and the control groups. The paternal educational level was somewhat higher among participants in the intervention group compared to the control group (P = 0.041).

Change in meal pattern

Table 2 shows an increased odds of consuming evening meals among participants in the intervention group than those in the control group (OR = 3.42 (95% CI: 1.21, 9.66)). No significant changes in consumption of breakfast, lunch, dinner, and frequency of family meals were reported as a result of the intervention.

Change in food and beverage availability and consumption

Table 3 indicates that the participants in the intervention group significantly decreased the availability of salty snacks than those in the control group ($\beta = -0.18$ (95% CI: -1.61, -0.15)). Furthermore, Table 4 shows a significant decrease in the reported daily consumption of salty snacks in the participants in the intervention group than those in the control group ($\beta = -0.18$ (95% CI: -7.69, -0.07)). A significant increase in daily water consumption was also reported by the intervention group compared to the control group ($\beta = 0.20$ (95% CI: 17.91, 466.42)). We revealed no other significant changes in availability or consumption of different food groups or beverages during the intervention period.

Change in nutrient intake and BMI z-score

According to results presented in Table 5, the intervention resulted in a significant decrease in total daily energy intake (kJ) in the intervention group compared to the control group ($\beta = -0.17$ (95% CI -2,131, -123)). Compared to the control group, the intervention group also significantly decreased the total intake of carbohydrates ($\beta = -0.17$ (95% CI -61.97, -6.23)), including mono- and disaccharides ($\beta = -0.21$ (95% CI -42.13, -7.29)) and sucrose ($\beta = -0.24$ (95% CI -21.26, -4.73)), and the intake of SFAs ($\beta = -0.17$ (95% CI -7.95, -0.11)). Finally, results showed an overall decrease in BMI and BMI *z*-score after the 6-month intervention in the intervention group compared to the control group ($\beta = -0.50$ (95% CI -1.25, -0.39) and $\beta = -0.21$ (95% CI -0.25, -0.10), respectively). See Table 5 for more information about results.

| Availability of food and beverage | | Baseline | | 6 mo | nths of follow-up | | Withi | n-group mean chang | es after 6 months | |
|-----------------------------------|-----------------------|------------------|-------|-----------------------|-------------------|-------|-----------------------|--------------------|----------------------|-------|
| | Intervention group | Control group | Å | Intervention group | Control group | ă | Intervention group | Control group | eta (95% CI) | ₩. |
| Sugar-sweetened soda (days/week) | 1.3 (0.8, 1.8) | 1.6 (1.0, 2.2) | 0.183 | 1.1 (0.6, 1.6) | 1.6 (1.0, 2.3) | 0.084 | -0.2 (-0.7, 0.3) | 0.17 (-0.3, 0.6) | -0.08 (-0.99, 0.27) | 0.261 |
| Diet soda (days/week) | 1.5 (0.9, 2.0) | 2.0 (1.2, 2.7) | 0.558 | 1.3 (0.8, 1.7) | 1.8 (1.1, 2.4) | 0.561 | -0.9 (-1.6, -0.2) | -0.9 (-1.6, -0.1) | -0.05 (-0.95, 0.48) | 0.516 |
| Fruit and vegetables (days/week) | 6.2 (5.7, 6.6) | 6.4 (6.0, 6.8) | 0.525 | 6.4 (6.1, 6.8) | 6.3 (5.8, 6.8) | 0.736 | 0.0 (-0.2, 0.3) | 0.3 (-0.2, 0.7) | 0.07 (-0.40, 0.84) | 0.476 |
| Sweets (days/week) | 2.5 (1.9, 3.1) | 2.6 (2.0, 3.2) | 0.420 | 2.1 (1.6, 2.7) | 2.7 (2.0, 3.4) | 0.314 | -0.4 (-1.0, 0.2) | 0.3 (-0.3, 0.9) | -0.13 (-1.41,0.15) | 0.110 |
| Salty snacks (days/week) | 2.3 (1.7, 2.9) | 2.0 (1.4, 2.6) | 0.850 | 1.7 (1.2, 2.2) | 2.3 (1.7, 3.0) | 0.191 | -0.6 (-1.1, -0.1) | 0.5 (-0.1, 1.1) | -0.18 (-1.61, -0.15) | 0.019 |

Intervention effect analyzed as difference in mean scores at baseline and after 6 months of follow-up in each variable (of each group) adjusted for baseline scores, age, sex, maternal-, and paternal educational intervention group and the control group, analyzed Differences in food and beverage availability between the

CI: confidence intervals; OR: odds ratio. models evel using linear regression

considered as statistically significant. was b-value < 0.05

Identification of intervention effects using multiple-group SEM of latent means

Pathways analyses, presented in Table 6, show that the intervention directly predicted BMI z-score ($\beta = -0.17$), total carbohydrates ($\beta = -0.20$), mono- and disaccharides $(\beta = -0.26)$, and sugar $(\beta = -0.22)$ after intervention. In addition, post-treatment levels of energy ($\beta = -0.65$), SFAs ($\beta = 0.43$), and total carbohydrates ($\beta = 0.41$) were directly associated with BMI z-score after intervention. Age and sex were directly linked to energy, mono- and disaccharides, and baseline BMI z-score. Age and sex indirectly predicted BMI z-score after intervention via energy, SFAs, and baseline BMI z-score (Table 6).

Discussion

The present study assessed the effects of a complex intervention program on meal pattern, food and beverage availability and consumption, nutrient intake, and weight status among children with overweight/obesity. Overall, the intervention successfully increased frequency of evening meals, decreased availability and consumption of salty snacks, and increased consumption of water in the intervention group compared to the control group after the intervention period. The main finding of the present study, which seemed to explain the observed decrease in BMI z-score during the intervention period, was a decrease in total energy intake, total carbohydrate intake, intake of mono- and disaccharides and sucrose, and intake of SFAs during the intervention period among participants in the intervention group compared to the control group.

Baseline results from the intervention and control group highlight the potential of reducing energy intake to improve weight status, as both groups confirmed a higher mean intake of total energy intake, total protein and carbohydrate intake, total intake of fat, intake of SFAs and cholesterol, and a similar intake of added sugar compared to estimates from the national dietary survey among 9-year-old Norwegian children (43).

The reduction in BMI z-score reported in the present study is in line with the effect size reported by other intervention studies focusing on behavior change among overweight and obese 6-11-year-olds, but results from previous studies reveal large variations in mean effects (from 0.1 to -0.71) (27). The reasons for the variations in effect could be due to differences between the intervention elements, age of participants, and setting. Further research has been warranted to identify optimal methods and strategies for weight management among children with overweight/obesity, including degree of parental involvement and how health practitioners can work with parents as agents of change in order to support health behaviors in their children (27, 30).

| Intake | | Baseline | | 6 mont | hs of follow-up | | With | in-group mean changes | after 6 months | |
|--|--|--|-----------------------|--|---|------------------------|---|---------------------------------|---------------------------|--------|
| | Intervention group | Control group | ă. | Intervention group | Control group | å | Intervention group | Control group | β (95% CI) | # |
| Water (g/d) | 612.1 (562.3, 661.8) | 554.0 (490.4, 617.5) | 0.153 | 659.5 (623.4, 695.6) | 576.7 (524.5, 628.8) | 0.010 | 47.4 (5.3, 89.5) | 22.7 (–34.1, 79.5) | 0.20 (17.91, 466.42) | 0.010 |
| Sugar-sweetened soda (g/d) | 154.0 (121.5, 250.6) | 186.1 (121.5, 250.6) | 0.439 | 85.0 (54.8, 115.1) | 130.8 (81.5, 180.2) | 0.116 | -69 (-119.4, 18.7) | -55.3 (-100.1, -10.4) | -0.11 (-82.58, 14.57) | 0.168 |
| Diet soda (g/d) | 34.2 (19.2, 49.1) | 38.4 (20.9, 55.9) | 0.713 | 39.6 (21.9, 57.3) | 26.1 (13.3, 39.0) | 0.220 | 5.4 (-7.4, 18.3) | -12.3 (-24.9, 0.3) | 0.11 (-2.18, 31.16) | 0.088 |
| Fruit and berries (g/d) | 150.0 (127.9, 172.8) | 156.1 (126.8, 185.3) | 0.758 | 134.7 (118.0, 151.5) | 149.8 (119.8, 179.8) | 0.383 | -15.6 (-35.0, 3.8) | -6.2 (-39.2, 26.7) | -0.03 (-35.97, 24.42) | 0.706 |
| Vegetables (g/d) | 94.0 (79.4, 108.5) | 97.5 (81.5, 113.5) | 0.745 | 96.5 (81.9, 111.0) | 99.9 (82.2, 117.6) | 0.762 | 2.5 (–8.6, 13.6) | 2.4 (-9.0, 13.9) | 0.03 (-12.38, 19.32) | 0.665 |
| White bread (g/d) | 6.6 (1.8, 11.3) | 7.0 ± (1.4, 12.5) | 0.915 | 6.6 (2.5, 10.8) | 6.4 (0.8, 12.0) | 0.956 | 0.0 (-6.1, 6.2) | -0.5 (-8.4, 7.3) | -0.01 (-7.69, 6.56) | 0.876 |
| Whole grain bread (g/d) | 138.3 (121.9, 154.8) | 126.8 (108.0, 145.6) | 0.357 | 122.0 (105.7, 138.2) | 123.4 (103.2, 143.6) | 0.911 | –16.4 (–35.5, 2.8) | –3.4 (–25.1, 18.4) | -0.02 (-27.99, 23.62) | 0.867 |
| Low-fat dairy products (g/d) | 257.1 (198.7, 307.7) | 250.8 (193.9, 307.7) | 0.879 | 268.7 (217.4, 320.1) | 306.9 (236.8, 377.1) | 0.382 | 11.7 (_41.2, 64.5) | 56.1 (121.1, 36.4) | -0.07 (-108.73, 44.68) | 0.410 |
| High-fat dairy products (g/d) | 62.5 (34.8, 90.2) | 80.6 (38.1, 123.1) | 0.477 | 50.2 (20.5, 79.9) | 110.6 (49.0, 172.1) | 0.081 | -12.3 (-46.8, 22.2) | 30.0 (–26.2, 86.1) | -0.13 (-113.09, 12.94) | 0.118 |
| Lean meat (g/d) | 30.1 (25.7, 34.5) | 31.2 (25.6, 36.8) | 0.753 | 29.8 (24.9, 34.8) | 33.7 (26.9, 40.5) | 0.364 | -0.3 (-5.2, 4.7) | 2.5 (-2.3, 7.3) | -0.07 (-10.13, 3.85) | 0.376 |
| Red/processed meat (g/d) | 59.0 (51.0, 67.0) | 56.9 (49.6, 64.2) | 0.705 | 54.5 (46.8, 62.2) | 54.9 (47.7, 62.0) | 0.951 | -4.5 (-12.4, 3.5) | -2.1 (-9.7,5.5) | -0.03 (-11.99, 7.98) | 0.691 |
| Sweets (g/d) | 26.2 (21.6, 30.9) | 19.6 (17.3, 21.9) | 0.012 | 19.8 (17.4, 22.3) | 23.8 (19.3, 28.2) | 0.127 | -6.4 (-11.2, -1.6) | 4.1 (-0.6, 8.9) | -0.14 (-8.95, 1.33) | 0.145 |
| Salty snacks (g/d) | 17.2 (14.2, 20.2) | 15.2 (12.4, 18.0) | 0.342 | 13.1 (10.9, 15.4) | 16.9 (13.6, 20.3) | 0.066 | -4.0 (-7.1,-1.0) | 1.7 (–2.4, 5.8) | -0.18 (-7.69, -0.07) | 0.046 |
| Data are presented *Differences in cons #Intervention effect a | as mean (95% CI). umption of food and l analyzed as difference | beverages between the in mean scores at base | interver eline and | ntion group and the con after 6 months of follo | ntrol group, analyzed us w-up in each variable (| ing indep of each g | vendent sample t-test (roup) adjusted for ba | s. aseline scores, age, sex, | maternal-, and paternal e | educa- |

Table 4. Changes in consumption of food and beverages during the 6-month intervention period

tional level using linear regression models. CI: confidence intervals. p-value < 0.05 was considered as statistically significant.

| | | Baseline | | 6 mont | chs of follow-up | | With | iin-group mean cha | nges after 6 months | |
|---|---|-------------------------|---------|------------------------|----------------------|----------|----------------------|------------------------|----------------------------------|----------|
| | Intervention group | Control group | å | Intervention group | Control group | å | Intervention group | Control group | β (95% CI) | 乱 |
| BMI (kg/m ²) | 23.8 (23.0, 24.6) | 23.1 (22.4, 23.9) | 0.129 | 24.0 (23.1,24.8) | 24.1 (23.3, 24.8) | 0.860 | 0.2 (-0.1, 0.4) | 0.9 (0.6, 1.2) | -0.50 (-1.25,-0.39) | <0.001 |
| BMI z-score | 2.04 (1.93, 2.15) | 2.02 (1.92, 2.13) | 0.825 | 1.93 (1.82, 2.04) | 2.07 (1.97, 2.16) | 0.06 | -0.11 (-0.16, -0.06) | 0.04 (0.00, 0.09) | -0.21 (-0.25,-0.10) | <0.001 |
| Energy (kJ) | 8,096 (7,428, 8,763) | 8,159 (7,431,8,886) | 0.899 | 6,978 (6,424, 7,532) | 8,226 (7,164,9,288) | 0.032 | -1,118 (-1,575,-662) | 67 (-893, 1,027) | -0.17 (-2131.15, -123.01) | 0.028 |
| Total protein (g) | 83.3 (76.9, 89.7) | 83.7 (76.7, 90.8) | 0.919 | 76.9 (70.6,83.2) | 87.4 (76.7, 98.1) | 0.081 | -6.4 (-10.7, -2.1) | 3.7(-5.9, 13.2) | -0.14 (-19.99, 0.46) | 0.061 |
| Total carbohydrates (g) | 243.7 (221.9, 265.5) | 247.0 (223.1, 270.8) | 0.839 | 203.5 (186.7, 220.3) | 242.1 (211.2, 273.0) | 0.024 | -40.2 (-56.6, -23.8) | -4.9(-29.6,19.8) | -0.17 (-61.97, -6.23) | 0.017 |
| Starch (g) | 120.7 (110.5, 131.0) | 120.9 (109.2, 132.6) | 0.977 | 107.4 (98.5, 116.3) | 118.5 (104.3, 132.7) | 0.174 | -13.3 (-22.5,-4.1) | -2,4 (-14.3, 9.5) | -0.10 (-23.83, 4.65) | 0.185 |
| Mono- and | 118.6 (103.8, 133.5) | 121.4 (105.9) | 0.796 | 92.6 (82.8, 102.3) | 120.0 (101.1, 138.9) | 0.008 | -26.1 (-38.0,-14.1) | -1,4 (-17.4, 14.6) | -0.21 (-42.13, -7.29) | 0.006 |
| disaccharides (g) | | | | | | | | | | |
| Sucrose (g) | 51.1 (43.8, 58.5) | 51.7 (44.1, 59.2) | 0.917 | 35.9 (30.8, 41.0) | 49.5 (41.3, 57.7) | 0.004 | -15.3 (-22.3, -8.2) | -2,2 (-9.2, 4.8) | -0.24 (-21.26, -4.73) | 0.002 |
| Fiber (g/1,000MJ) | 2.9 (2.8, 3.0) | 2.9 (2.7, 3.0) | 0.627 | 3.0 (2.8, 3.1) | 2.8±(2.6, 3.0) | 0.141 | 0.1 (-0.1, 0.2) | -0,1 (-0.2, 0.1) | 0.14 (-0.03, 0.41) | 0.081 |
| Total fat (g) | 63.6 (57.4, 69.8) | 63.5 (57.2, 69.8) | 0.998 | 55.3 (50.4, 60.2) | 66.0 (56.0, 76.1) | 0.047 | -8.26(-12.7,-3.8) | 2.5(-8.0, 12.9) | -0.16 (-19.86, 0.66) | 0.066 |
| MUFAs (g) | 23.3 (20.9, 25.7) | 23.1 (20.7, 25.4) | 0.897 | 20.2 (18.3, 22.0) | 24.0 (20.3, 27.7) | 0.055 | -3. l (-4.8,-l .3) | 0.9(–3.1,4.9) | -0.15 (-7.25, 0.43) | 0.081 |
| SFAs (g) | 25.7 (23.2, 28.3) | 26.0 (23.4, 28.6) | 0.890 | 22.4 (20.2, 24.5) | 27.0 (23.2, 30.8) | 0.029 | -3.4 (-5.3,-1.4) | I.0 (–2.9, 4.9) | -0.17 (-7.95,-0.11) | 0.044 |
| PUFAs (g) | 8.3 (7.4, 9.3) | 8.2 (7.3, 9.2) | 0.896 | 7.4 (6.6, 8.2) | 8.6 (6.7,10.6) | 0.220 | -1.0(-1.6,-0.3) | 0.4(-1.6, 2.4) | -0.11 (-3.19,0.74) | 0.220 |
| n–3 fatty acids (g) | 0.67 (0.40, 0.93) | 0.54 (0.39, 0.69) | 0.436 | 0.69 (0.46,0.91) | 0.64 (0.35, 0.93) | 0.774 | 0.02 (-0.22, 0.26) | 0.10(-0.20,0.39) | 0.02 (-0.30, 0.40) | 0.789 |
| n–6 fatty acids (g) | 1.11 (0.83, 1.40) | 0.93 (0.74, 1.12) | 0.301 | 0.98 (0.72, 1.24) | 1.06 (0.66, 1.46) | 0.729 | -0.13 (-0.40, 0.13) | 0.13(-0.31,0.57) | -0.06 (-0.64, 0.30) | 0.478 |
| Cholesterol (mg) | 250.4 (222,8, 277.9) | 281.7 (234.8, 328.5) | 0.236 | 249.1 (212.6, 285.6) | 298.3 (232.8, 363.8) | 0.176 | -40.2 (-56.6, -23.8) | I 6.6 (–50.9, 84.2) | -0.08 (-99.95, 36.52) | 0.359 |
| E% protein | 17.8 (17.2, 18.3) | 17.6 (17.0, 18.2) | 0.698 | 18.7 (18.2, 19.2) | 18.2 (17.6, 18.9) | 0.228 | 1.0 (0.4, 1.5) | 0.6 (0.2, 1.1) | 0.06 (–0.43, 0.99) | 0.435 |
| E% Fat | 28.9 (27.8, 30.0) | 28.7 (27.6, 29.9) | 0.847 | 29.3 (28.2, 30.3) | 29.4 (28.3, 30.4) | 0.900 | 0.4 (-0.7, 1.5) | 0.6 (-0.5, 1.8) | 0.00 (–1.36, 1.40) | 0.980 |
| E% SFAs | 11.6 (11.2, 12.1) | 11.8 (11.2, 12.3) | 0.727 | 11.8 (11.8, 11.3) | 12.1 (11.5, 12.6) | 0.429 | 0.1 (-0.4, 0.7) | 0.3 (-0.3, 0.9) | -0.04 (-0.87, 0.56) | 0.662 |
| E% MUFAs | 10.6 (10.1, 11.1) | 10.4 (9.9, 10.9) | 0.669 | 10.7 (10.2, 11.2) | 10.7 (10.2, 11.2) | 0.987 | 0.1 (-0.4, 0.6) | 0.3 (-0.2, 0.8) | 0.00 (–0.61, 0.64) | 0.962 |
| E% PUFAs | 3.8 (3.5, 4.0) | 3.7 (3.5, 3.9) | 0.624 | 3.9 (3.6, 4.2) | 3.7 (3.5, 3.9) | 0.276 | 0.1 (-0.1, 0.3) | 0.0 (-0.2, 0.2) | 0.06 (–0.16, 0.42) | 0.360 |
| E% Carbohydrates | 51.0 (49.7, 52.3) | 51.4 (50.1, 52.6) | 0.706 | 49.6 (48.5,50.8) | 50.2 (48.9, 51.5) | 0.531 | -I.4 (-2.5, -0.2) | -I.2 (-2.4, 0.0) | -0.04 (-1.90, 1.07) | 0.579 |
| E% Sucrose | 10.4 (9.4, 11.3) | 10.7 (9.5, 11.8) | 0.686 | 8.6 (7.9, 9.5) | 10.1 (9.2, 11.1) | 0.021 | -1.7 (-2.7, -0.7) | -0.5 (-1.5, 0.5) | -0.19 (-2.55,-0.26) | 0.017 |
| Data are presented as m BMI: body mass index; E% | an (95% CI). : percentage of total ∈ | energy; g: grams: SFAs: | saturat | ed fatty acids; MUFAs: | : monounsaturated fa | ty acids | PUFAs: polyunsatura | ted fatty acids; kJ: k | ilojoule; g: grams; mg: milligra | tms; MJ: |

*Significant difference (P < 0.05) between the intervention group and the control group, analyzed using independent sample t-tests. #Significant difference (P < 0.05) between mean scores at baseline and after 6 months of follow-up in each variable (of each group) adjusted for baseline scores, age, sex, maternal-, and paternal educational megajoule.

level using linear regression models. p-value < 0.05 was considered as statistically significant.

Table 6. Direct and indirect effects of the parsimonious structural equation model

| Pathways | β | SE | Р |
|--|-------|------|--------|
| Direct effects | | | |
| Intervention \rightarrow Total carbohydrates | -0.20 | 0.09 | 0.026 |
| Intervention \rightarrow Mono- and disaccharides | -0.26 | 0.08 | 0.002 |
| Intervention \rightarrow Sucrose | -0.22 | 0.09 | 0.011 |
| Intervention \rightarrow BMI z-score post-intervention | -0.17 | 0.04 | <0.001 |
| Age ightarrow Energy | -0.23 | 0.09 | 0.007 |
| $Age \rightarrow SFAs$ | -0.20 | 0.09 | 0.024 |
| $Age \rightarrow Mono-$ and disaccharides | -0.13 | 0.06 | 0.040 |
| $Age \rightarrow BMI z$ -score baseline | -0.36 | 0.08 | <0.001 |
| $Sex \to Energy$ | 0.22 | 0.09 | 0.011 |
| $Sex \to SFAs$ | 0.18 | 0.08 | 0.008 |
| Sex \rightarrow Mono- and disaccharides | 0.19 | 0.09 | 0.023 |
| $Sex \rightarrow BMI z$ -score baseline | 0.34 | 0.07 | <0.001 |
| Energy \rightarrow BMI z-score post-intervention | -0.65 | 0.24 | 0.007 |
| $SFAs{\to}BMI\ z{\operatorname{-score}}\ post-intervention$ | 0.43 | 0.13 | 0.001 |
| Total carbohydrates \rightarrow BMI z-score post-intervention | 0.41 | 0.19 | 0.032 |
| BMI baseline \rightarrow BMI z-score post-intervention | 0.85 | 0.03 | <0.001 |
| Indirect effects | | | |
| Age \rightarrow BMI z-score post-intervention | -0.55 | 0.02 | <0.001 |
| Age \rightarrow Energy \rightarrow BMI z-score post-intervention | | | |
| $Age \to SFAs \to BMI \text{ z-score post-intervention}$ | | | |
| Age \rightarrow BMI z-score baseline \rightarrow BMI z-score post-intervention | | | |
| Sex \rightarrow BMI z-score post-intervention | -0.22 | 0.06 | <0.001 |
| $Sex \to Energy \to BMI \ z\text{-}score \ post-intervention$ | | | |
| $Sex \to SFAs \to BMI\ z\text{-score post-intervention}$ | | | |
| Sex \rightarrow BMI baseline \rightarrow BMI z-score post-intervention | | | |

BMI: body mass index; SFAs: saturated fatty acids; SE: standard error.

Another recent systematic review on family-based nutrition interventions for obesity prevention concluded that most successful intervention studies included a cluster of behavior change techniques, such as setting family-based goals, modifying home food environment, hands-on approaches to teaching nutrition (games and group-based activities), and fruit and vegetable vouchers (44). Using active family-based learning involving meal preparation and cooking has been associated with increased parent-child communication, teamwork, and inspiration to implement healthy eating as part of everyday life (45, 46), which might explain positive changes in nutrient intake observed in the present study. This conclusion is also supported by results from a recent systematic review, which presented experimental learning as a useful strategy to improve knowledge, attitudes, and behaviors toward healthy eating among children (47).

Unfortunately, due to the design of our study, we were unable to entangle the effects of parental involvement on outcomes in our study. However, in line with the program components used in our study, previous weight management studies among children with overweight/obesity have emphasized the importance of involving parent as the agent of change through nutrition education sessions, using an authoritative parenting style, enhancing positive parenting skills and child behavior management strategies (30, 48). Furthermore, a 12-month intervention study showed that allocating children with overweight/obesity to multiple-family intervention significantly decreased waist circumference, but not BMI, then those allocated into single-family interventions (49). However, no additional long-term effect of group vs individual family intervention in treatment of childhood obesity was shown at 36 months follow-up (50).

A systematic review of primary healthcare obesity interventions also indicated a positive relationship between hours of contact and treatment effect (51). This conclusion has been supported by previous findings of a study based on primary care screening for identification and follow-up by brief parent counseling (four consultations over a 12-week period) of children, aged 5–10 years old, with overweight and obesity, which did not result in improved BMI, physical activity, or nutrition among participants (52). Thus, the high frequency of individual consultations and group-based courses combined may most likely have contributed the positive treatment effects observed in the present study. Results from our study in combination with previously reported findings may therefore indicate that current policies in several countries that ensure use of national screening programs in combination with brief, individualized prevention measures by primary care sector to reduce childhood obesity should be reevaluated.

Our findings also identified that age and sex to be relevant determinants of nutrient intake and weight status once higher age and being female predicted lower intake macronutrients during the intervention period and consequently reduced BMI z-score. Relatively few studies have previously investigated sex-based differences in childhood obesity, but the importance of including sex-specific focus to increase effectiveness of future family-based interventions has been warranted (53). Previous studies have also shown that specific approaches may be important for some age groups, but not for others, depending on their level of self-regulation skills: i.e. setting family-based goal and including hands-on nutrition activities such as games seem to effectively contribute to change dietary habits among younger children, but not adolescents (54, 55). Findings from previously published studies among children and adolescents suggest that parental support as an influence on child health behaviors tends to peak at age 12 (56), which support the use of a family-based approach in the present study.

Strength and limitations of study

A limitation of the current controlled trial is the lack of a randomized design, and a time lag between data collection and publication. A recent systematic review has also confirmed that relatively few randomized controlled trials testing the efficacy of family-based nutrition interventions targeting school aged children with overweight/obesity have been published (44). On the other hand, the study design used in the present study was tailored and adapted for intervention in ordinary community-based settings, which may increase the likelihood of implementation in regular services. Another limitation of the study was that despite mandatory screening of height and weight among Norwegian preschoolers and 3rd graders, the recruitment process in regular services in several different municipalities did not allow us to control this process and ensure that all children and families in the target group had been invited to participate in the present study. Moreover, high dropout rates and low number of participants in each group who completed the study are other limitations that

should be mentioned. However, the inclusion of 143 participants yielded an adequate power of 94%. In line with our study, most previously published results from child obesity prevention trials are limited by low sample size (57). Moreover, as municipal healthcare institutions have a responsibility to prevent and treat overweight and obesity among all age groups (22), we used a waiting list design, which did not allow long-term controlled follow-up.

A strength of the present study was the use of a structural theory and evidence framework for studying development, implementation, and evaluation. Moreover, results from the pilot study provided information on opportunities and barriers to tailor the intervention to the local context in Norwegian municipalities and to adapt the intervention tools and materials to health personnel conducting the study and to the target group (31). Another strength was the use of objectively measured height and weight by trained public health nurses, and the use of a validated and comprehensive FFQ, which provided estimates of nutrient intake (37). Moreover, the use of structural equation modeling as a statistical approach for the evaluation of multiple relationships between variables provided increased knowledge about pathways of change.

Future perspectives

Considering that overweight and obesity among children is a global problem and to limit stains on the healthcare system, future interventions should reevaluate strategies used to conduct family-based treatment programs targeting children with overweight/obesity. Emerging evidence indicates that the incorporation of digital technology together with conventional treatments may be an acceptable approach to increase effectiveness in weight management intervention studies (58). In addition, there is a need to evaluate possible long-term effects and how interventions effects might vary among different sub-groups, such as non-traditional families, and racial and ethnic minorities.

Conclusion

A positive intervention effect was observed on evening meal pattern, availability and consumption of salty snacks, and consumption of water. Moreover, estimates indicating a decreased total intake of energy and carbohydrates, including total carbohydrate intake, intake of mono- and disaccharides and sucrose, and intake of SFAs seem to explain the observed decrease in BMI *z*-score during the intervention period. Thus, the creation of environments supportive of healthier behaviors seems to mitigate challenges related to childhood overweight and obesity.

Acknowledgments

We are grateful to all the families who took part in the pilot study, to public health nurses and other healthcare providers at Public Health Clinics and Healthy Life Centers for collaborating with us for the development of the intervention, and to Eivind Meland for his comments on this article. This study was financed by the Research Council of Norway (NFR) (Grant Number 228454). As a project leader, Eivind Meland received the funding.

Conflict of interest and funding

The authors declare no potential conflicts of interest. The funders had no role in the study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Ethics and consent

The Regional Committee for Medical and Health Research Ethics approved the study (no. 2013/1291), which was conducted in accordance with the Helsinki Declaration of 1975, as revised in 2008. The children and their parents received oral and written information about the study, and parents signed a written consent form prior to participation.

References

- NCD Risk Factor Collaboration (NCD-RisC). Worldwide trends in body-mass index, underweight, overweight, and obesity from 1975 to 2016: a pooled analysis of 2416 population-based measurement studies in 128-9 million children, adolescents, and adults. Lancet 2017; 90(10113): 2627–42. doi: 10.1016/S0140-6736(17)32129-3
- Garrido-Miguel M, Cavero-Redondo I, Álvarez-Bueno C, Rodríguez-Artalejo F, Moreno LA, Ruiz JR, et al. Prevalence and trends of overweight and obesity in European children from 1999 to 2016: a systematic review and meta-analysis. JAMA Pediatr 2019; 173(10): e192430. doi: 10.1001/ jamapediatrics.2019.2430
- Øvrebø B, Bergh IH, Stea TH, Bere E, Surén P, Magnus PM, et al. Overweight, obesity, and thinness among a nationally representative sample of Norwegian adolescents and changes from childhood: associations with sex, region, and population density. PLoS One 2021; 16(8): e0255699. doi: 10.1371/journal.pone.0255699
- Umer A, Kelley GA, Cottrell LE, Giacobbi P, Jr, Innes KE, Lilly CL. Childhood obesity and adult cardiovascular disease risk factors: a systematic review with meta-analysis. BMC Public Health 2017; 17(1): 683. doi: 10.1186/s12889-017-4691-z
- Simmonds M, Llewellyn A, Owen CG, Woolacott N. Predicting adult obesity from childhood obesity: a systematic review and meta-analysis. Obes Rev 2016; 17(2): 95–107. doi: 10.1111/ obr.12334
- Llewellyn A, Simmonds M, Owen CG, Woolacott N. Childhood obesity as a predictor of morbidity in adulthood: a systematic review and meta-analysis. Obes Rev 2016; 17(1): 56–67. doi: 10.1111/obr.12316
- Farpour-Lambert NJ, Baker JL, Hassapidou M, Holm JC, Nowicka P, O'Malley G, et al. Childhood obesity is a chronic disease demanding specific health care – a position statement from the Childhood Obesity Task Force (COTF) of the

Citation: Food & Nutrition Research 2023, 67: 9617 - http://dx.doi.org/10.29219/fnr:v67.9617

European Association for the Study of Obesity (EASO). Obes Facts 2015; 8(5): 342–9. doi: 10.1159/000441483

- Daniels SR, Arnett DK, Eckel RH, Gidding SS, Hayman LL, Kumanyika S, et al. Overweight in children and adolescents: pathophysiology, consequences, prevention, and treatment. Circulation 2005; 111(15): 1999–2012. doi: 10.1161/01. CIR.0000161369.71722.10
- 9. Karnik S, Kanekar A. Childhood obesity: a global public health crisis. Int J Prev Med 2012; 3(1): 1–7.
- Thomas-Eapen N. Childhood obesity. Prim Care 2021; 48(3): 505–15. doi: 10.1016/j.pop.2021.04.002
- Davison KK, Birch LL. Childhood overweight: a contextual model and recommendations for future research. Obesity Rev 2001; 2(3): 159–71. doi: 10.1046/j.1467-789x.2001.00036.x
- Grydeland M, Bergh IH, Bjelland M, Lien N, Andersen LF, Ommundsen Y, et al. Correlates of weight status among Norwegian 11-year-olds: the HEIA study. BMC Public Health 2012; 12: 1053. doi: 10.1186/1471-2458-12-1053
- MacFarlane A, Cleland V, Crawford D, Campbell K, Timperio A. Longitudinal examination of the family food environment and weight status among children. Int J Pediatr Obes 2009; 4(4): 343–52. doi: 10.3109/17477160902846211
- Buoncristiano M, Williams J, Simmonds P, Nurk E, Ahrens W, Nardone P, et al. Socioeconomic inequalities in overweight and obesity among 6- to 9-year-old children in 24 countries from the World Health Organization European region. Obes Rev 2021; 22(Suppl. 6): S13213. doi: 10.1111/obr.13213
- Bridger Staatz C, Kelly Y, Lacey RE, Blodgett JM, George A, Arnot M, et al. Socioeconomic position and body composition in childhood in high- and middle-income countries: a systematic review and narrative synthesis. Int J Obes (Lond) 2021; 45(11): 2316–34. doi: 10.1038/s41366-021-00899-y
- Darmon N, Drewnowski A. Does social class predict diet quality? Am J Clin Nutr 2008; 87: 1107–17. doi: 10.1093/ ajcn/87.5.1107
- 17. Moschonis G, Tanagra S, Vandorou A, Kyriakou AE, Dede V, Siatitsa PE, et al. Social, economic and demographic correlates of overweight and obesity in primary-school children: preliminary data from the Healthy Growth Study. Public Health Nutr 2010; 13: 1693–700. doi: 10.1017/S1368980010002247
- Manios Y, Androutsos O, Katsarou C, Vampouli EA, Kulaga Z, Gurzkowska B, et al. Prevalence and sociodemographic correlates of overweight and obesity in a large Pan-European cohort of preschool children and their families: the ToyBox study. Nutrition 2018; 55–56: 192–8. doi: 10.1016/j.nut.2018.05.007
- Mekonnen T, Papadopoulou E, Arah OA, Brantsæter AL, Lien N, Gebremariam MK. Socioeconomic inequalities in children's weight, height and BMI trajectories in Norway. Sci Rep 2021; 11: 4979. doi: 10.1038/s41598-021-84615-w
- Ayine P, Selvaraju V, Venkatapoorna CMK, Bao Y, Gaillard P, Geetha T. Eating behaviors in relation to child weight status and maternal education. Children (Basel) 2021; 8(1): 32. doi: 10.3390/children8010032
- Jebeile H, Kelly AS, O'Malley G, Baur LA. Obesity in children and adolescents: epidemiology, causes, assessment, and management. Lancet Diabetes Endocrinol 2022; 10(5): 351–65. doi: 10.1016/S2213-8587(22)00047-X
- 22. The Norwegian Directorate of Health. Nasjonale faglige retningslinjer for primærhelsetjenesten. Forebygging og behandling av overvekt og fedme hos barn og unge (National guideline for the prevention, identification and treatment of overweight and obesity in children and adolescents). Oslo: The Norwegian Directorate of Health; 2010.

- 23. The Norwegian Directorate of Health. Nasjonale faglige retningslinjer for veiing og måling i helsestasjons-og skolehelsetjenesten (National guideline for weighing and measuring at health clinics and in the school health service). Oslo: The Norwegian Directorate of Health; 2010.
- 24. Westwood M, Fayter D, Hartley S, Rithalia A, Butler G, Glasziou P, et al. Childhood obesity: should primary school children be routinely screened? A systematic review and discussion of the evidence. Arch Dis Child 2007; 92(5): 416–22. doi: 10.1136/adc.2006.113589
- Pereira AR, Oliveira A. Dietary interventions to prevent childhood obesity: a literature review. Nutrients 2021; 13(10): 3447. doi: 10.3390/nu13103447
- Elvsaas I-KØ, Juvet LK, Giske L, Fure B. Effectiveness of interventions for overweight or obesity in children and adolescents. Oslo: The Norwegian Institute of Public Health; 2016.
- 27. Mead E, Brown T, Rees K, Azevedo LB, Whittaker V, Jones D, et al. Diet, physical activity and behavioural interventions for the treatment of overweight or obese children from the age of 6 to 11 years. Cochrane Database Syst Rev 2017; 6(6): CD012651. doi: 10.1002/14651858.CD012651
- Bronfenbrenner U. The ecology of human development: experiments by nature and design. Cambridge, MA: Harvard University Press; 1979.
- Golan M. Parents as agents of change in childhood obesityfrom research to practice. Int J Pediatr Obes 2006; 1(2): 66–76. doi: 10.1080/17477160600644272
- Chai LK, Collins C, May C, Brain K, Wong See D, Burrows T. Effectiveness of family-based weight management interventions for children with overweight and obesity: an umbrella review. JBI Database System Rev Implement Rep 2019; 17(7): 1341– 427. doi: 10.11124/JBISRIR-2016-003082
- 31. Stea TH, Haugen T, Berntsen S, Guttormsen V, Øverby NC, Haraldstad K, et al. Using the intervention mapping protocol to develop a family-based intervention for improving lifestyle habits among overweight and obese children: study protocol for a quasi-experimental trial. BMC Public Health 2016; 16(1): 1092. doi: 10.1186/s12889-016-3766-6
- 32. Cole TJ, Lobstein T. Extended international (IOTF) body mass index cut-offs for thinness, overweight and obesity. Pediatr Obes 2012; 7(4): 284–94. doi: 10.1111/j.2047-6310.2012.00064.x
- Berntsen S, Mowinckel P, Carlsen KH, Lødrup Carlsen KC, Pollestad Kolsgaard ML, Joner G, et al. Obese children playing towards an active lifestyle. Int J Pediatr Obes 2010; 5(1): 64–71. doi: 10.3109/17477160902957166
- 34. Faul F, Erdfelder E, Buchner A, Lang AG. Statistical power analyses using G*Power 3.1: tests for correlation and regression analyses. Behav Res Methods 2009; 41(4): 1149–60. doi: 10.3758/ BRM.41.4.1149
- 35. Ryan RM, Deci EL. Active human natre: self-determination theory and the promotion and maintenance of sport, exercise, and health. In: Haggar MS, Chatzisarantis NLD, eds. Intrinsic motivation and self-determination in exercise and sport. Leeds: Human Kinetics Europe; 2007, pp. 1–19.
- Miller WR. Motivational interviewing: research, practice, and puzzles. Addict Behav 1996; 21(6): 835–42. doi: 10.1016/0306-4603(96)00044-5
- 37. Overby NC, Johannesen E, Jensen G, Skjaevesland AK, Haugen M. Test-retest reliability and validity of a web-based food-frequency questionnaire for adolescents aged 13–14 to be used in the Norwegian Mother and Child Cohort Study (MoBa). Food Nutr Res 2014; 58(1): 23956. doi: 10.3402/fnr. v58.23956

- Blaker B, Aarsland M. Mål og vekt for matvarer (The volume and weight for food items). 2nd ed. Oslo: Falch hurtigtrykk; 1995.
- Lauritsen J. FoodCalc v.1.3. 2021. Available from: https://github. com/jesperldk/FoodCalc [cited 3 March 2023].
- The Norwegian Food Safety Authority. The Norwegian food composition table. 2021. Available from: https://www.matvaretabellen.no/?language=en [cited 3 March 2023].
- Shrewsbury V, Wardle J. Socioeconomic status and adiposity in childhood: a systematic review of cross-sectional studies 1990– 2005. Obesity (Silver Spring) 2008; 16(2): 275–84. doi: 10.1038/ oby.2007.35
- Hu LT, Bentler PM. Cutoff criteria for fit indexes in covariance structure analysis: conventional criteria versus new alternatives. Struct Equ Model Multidiscip J 1999; 6(1): 1–55. doi: 10.1080/10705519909540118
- 43. Hansen LB, Myhre JB, Johansen AMW, Paulsen MM, Andersen LF. UNGKOST 3, Nationwide dietary survey among pupils in 4th and 8th grade in Norway, 2015. Oslo: The Norwegian Directorate of Health; 2016.
- Perdew M, Liu S, Naylor PJ. Family-based nutrition interventions for obesity prevention among school-aged children: a systematic review. Transl Behav Med 2021; 11(3): 709–23. doi: 10.1093/tbm/ibaa082
- Berge JM, MacLehose RF, Larson N, Laska M, Neumark-Sztainer D. Family food preparation and its effects on adolescent dietary quality and eating patterns. J Adolesc Health 2016; 59(5): 530–6. doi: 10.1016/j.jadohealth.2016.06.007
- van der Horst K, Ferrage A, Rytz A. Involving children in meal preparation. Effects on food intake. Appetite 2014; 79: 18–24. doi: 10.1016/j.appet.2014.03.030
- 47. Varman SD, Cliff DP, Jones RA, Hammersley ML, Zhang Z, Charlton K, et al. Experiential learning interventions and healthy eating outcomes in children: a systematic literature review. Int J Environ Res Public Health 2021; 18(20): 10824. doi: 10.3390/ijerph182010824
- 48. Kitzman-Ulrich H, Wilson DK, St George SM, Lawman H, Segal M, Fairchild A. The integration of a family systems approach for understanding youth obesity, physical activity, and dietary programs. Clin Child Fam Psychol Rev 2010; 13(3): 231–53. doi: 10.1007/s10567-010-0073-0
- Kokkvoll A, Grimsgaard S, Ødegaard R, Flægstad T, Njølstad I. Single versus multiple-family intervention in childhood overweight- Finnmark Activity School: a randomised trial. Arch Dis Child 2014; 99(3): 225–31. doi: 10.1136/ archdischild-2012-303571
- Kokkvoll AS, Grimsgaard S, Flaegstad T, Andersen LB, Ball GDC, Wilsgaard T, et al. No additional long-term effect of group vs individual family intervention in the treatment of childhood obesity – a randomised trial. Acta Paediatr 2020; 109(1): 183–92. doi: 10.1111/apa.14916
- Whitlock EP, O'Connor EA, Williams SB, Beil TL, Lutz KW. Effectiveness of weight management interventions in children: a targeted systematic review for the USPSTF. Pediatrics 2010; 125(2): e396–418. doi: 10.1542/peds.2009-1955
- 52. Wake M, Baur LA, Gerner B, Gibbons K, Gold L, Gunn J, et al. Outcomes and costs of primary care surveillance and intervention for overweight or obese children: the LEAP 2 randomised controlled trial. BMJ 2009; 339: b3308. doi: 10.1136/ bmj.b3308
- 53. Shah B, Tombeau Cost K, Fuller A, Birken CS, Anderson LN. Sex and gender differences in childhood obesity: contributing to the research agenda. BMJ Nutr Prev Health 2020; 3(2): 387–90. doi: 10.1136/bmjnph-2020-000074

- Pandita A, Sharma D, Pandita D, Pawar S, Tariq M, Kaul A. Childhood obesity: prevention is better than cure. Diabetes Metab Syndr Obes 2016; 9: 83–9. doi: 10.2147/DMSO.S90783
- 55. Suleiman-Martos N, García-Lara RA, Martos-Cabrera MB, Albendín-García L, Romero-Béjar JL, Cañadas-De la Fuente GA, et al. Gamification for the improvement of diet, nutritional habits, and body composition in children and adolescents: a systematic review and meta-analysis. Nutrients 2021; 13(7): 2478. doi: 10.3390/nu13072478
- Rhodes RE, Lim C. Promoting parent and child physical activity together: elicitation of potential intervention targets and preferences. Health Educ Behav 2018; 45(1): 112–23. doi: 10.1177/1090198117704266
- 57. Brown T, Moore TH, Hooper L, Gao Y, Zayegh A, Ijaz S, et al. Interventions for preventing obesity in children. Cochrane

Database Syst Rev 2019; 7(7): CD001871. doi: 10.1002/14651858. CD001871.pub4

 Kouvari M, Karipidou M, Tsiampalis T, Mamalaki E, Poulimeneas D, Bathrellou E, et al. Digital health interventions for weight management in children and adolescents: systematic review and meta-analysis. J Med Internet Res 2022; 24(2): e30675. doi: 10.2196/30675

*Tonje Holte Stea

Department of Health and Nursing Science, University of Agder, Postbox 422 4604 Kristiansand, Norway. E-mail: tonje.h.stea@uia.no